Author's response to reviews

Title: Report of a case of sciatica due to extra-pelvic heterotopic ossification.

Authors:

Elias C Panagiotopoulos (ecpanagi@med.upatras.gr)
Spyros A Syggelos (sasyggelos@gmail.com)
Athanasios Plotas (plotas@tpp24.gr)
Gregorios Tsigkas (gregtsig@mail.gr)
Panagiotis Dimopoulos (pdim@med.upatras.gr)

Version: 4 Date: 23 April 2008

Author's response to reviews:

To the JMCR Editorial Team,

Dear Sirs,

I’m sending you a revised version of our manuscript ‘REPORT OF A CASE OF SCIATICA DUE TO EXTRA-PELVIC HETEROTOPIC OSSIFICATION’.

We have tried to improve the English language. We have also tried to make the neurological findings more clear. We have also changed the format of references according to journal’s guidelines.

If this is not enough, we would like to have, if possible, one more chance.

On behalf of the Authors

Spyros Syggelos